

G OPEN ACCESS

Citation: Takada T, Kitajima T (2022) Trendfollowing with better adaptation to large downside risks. PLoS ONE 17(10): e0276322. <u>https://doi.org/</u> 10.1371/journal.pone.0276322

Editor: Ning Du, DePaul University, UNITED STATES

Received: April 16, 2022

Accepted: October 4, 2022

Published: October 18, 2022

Copyright: © 2022 Takada, Kitajima. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results presented in this study are the four daily stock indices for 47 years which are retrieved from Datastream database. The data are owned by a third-party organization, Refinitiv. The authors were able to access the data via Graduate School of Business, Osaka Metropolitan University that concludes a license contract with Refinitiv Japan. The authors did not have any special privileges in accessing the data that others would not have. Any researchers wishing to access the Datastream database may do so by contracting with Refinitiv (https://www.refinitiv.com/en/contact-us). The time RESEARCH ARTICLE

Trend-following with better adaptation to large downside risks

Teruko Takada ^{1,2}, Takahiro Kitajima^{1,2}

1 Graduate School of Business, Osaka Metropolitan University, Osaka, Japan, 2 Faculty of Commerce, Kumamoto Gakuen University, Kumamoto, Japan

* ttakada@omu.ac.jp

Abstract

Avoiding losses from long-term trend reversals is challenging, and trend-following is one of the few trading approaches to explore it. While trend-following is popular among investors and has gained increased attention in academia, the recent diminished profitability in equity markets casts doubt on its effectiveness. To clarify its cause and suggest remedies, we thoroughly examine the effect of market conditions and averaging window on recent profitability using four major stock indices in an out-of-sample experiment comparing trend-following rules (moving average and momentum) and a machine-classification-based non-trend-following rule. In addition to the significant advantage of trend-following rules in avoiding downside risks, we find a discrepancy in the optimum averaging window size between trend direction phases, which is exacerbated by a higher positive trend direction ratio. A higher positive trend direction ratio leads to poor performance relative to buy-and-hold returns. This discrepancy creates the dilemma of choosing which trend direction phase to emphasize. Incorporating machine-learning into trend-following is effective for alleviating this dilemma. We find that the profit-maximizing averaging window realizes the level that best balances the dilemma and suggest a simple guideline for selecting the optimum averaging window. We attribute the sluggishness of trend-following in recent equity markets to the insufficient chances of trend reversals rather than their loss of profitability. Our results contribute to improving the performance of trend following by mitigating the dilemma.

1 Introduction

Stock markets have recurring upward and downward trend direction phases, and investors can lose much of the earned profit during a maximum drawdown. However, avoiding losses from trend reversals is a challenging task involving fundamental difficulties. This study considers the problem of long-term trend direction classification to help avoid losses from such trend reversals.

The essential difficulty of the long-term trend direction classification problem lies in the distinction between noisy daily stock price fluctuations and the rare large price trend direction reversals. Given the tendency for large, sudden trend reversals in the stock market after a long, stable, and continued upward trend, we can interpret the difficulty for investors as a

series plot of all raw data this study used are presented in Fig 2.

Funding: This work was supported by JSPS (Japan Society for the Promotion of Science) KAKENHI Grant Number 21K18439 to TT; https://www.jsps. go.jp/english/e-grants/. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

decision between "keep riding the upward long-term trend" or "be cautious about sudden trend reversals".

This situation corresponds to the *stability-plasticity dilemma* [1] that both human and learning machines often encounter in the process of knowledge acquisition when confronted with new environments, typically involving the problem of non-stationarity (or concept drift) with class imbalance. Non-stationarity and class imbalance constitute a combination representing the most challenging current problems for computational intelligence studies [2]: the machine classification problem is complicated by non-stationarity, where the underlying data distribution is time-varying and insufficient information on the minority class complicates it further.

In particular, dealing with non-stationarity is highly challenging [3]. Several studies propose remedies to address non-stationarity by classifying the types of non-stationarity. However, they consist mainly of general-purpose theory-oriented approaches [4–6] are critical of their practical performance. They emphasize the importance of simplicity in the model by tailoring it to each specific application problem. Because of such difficulties dealing with non-stationarity, trend direction classification studies using machine classification focus on the short term, with temporal horizons of daily and monthly at most.

The handling of simultaneous concept drift and class imbalance is attracting increased attention, especially in real-world online learning. However, it is still an emerging research area, and very few studies address the simultaneous occurrence of these problems [7]. Applying the proposed solutions to real-world data is often difficult because the expected characteristics of the data are usually unknown [8].

As a trading strategy, *trend-following* is possibly the only approach that addresses the *long-term* trend direction classification problem while providing the desired simplicity. Trend-following is a form of technical analysis that aims to ride an upward trend formation while avoiding losses by identifying the downward trend reversals in time. Trend-following rules generate buy (sell) signals when users identify the start (end) of a trend formation. In the simplest formulation, a single parameter, the *averaging window*, also called the *look-back period*, balances "the stability of the long-term trend" and the "sensitivity (plasticity) to trend direction changes". That is, the dilemma in choosing to emphasize the "clear capture of an upward trend" and "sensitivity to trend reversal signals". Hence, selecting the appropriate averaging window becomes the key to improving trading performance.

This study aims to empirically clarify the source of the recent unstable and diminished returns of trend-following in equity markets and suggest remedies to realize their improved advantage in avoiding long-term trend reversals. We conduct the following two studies. First, we perform a comprehensive out-of-sample comparative analysis that focuses on the effect of trend direction and strength on profitability in relation to the averaging window size. We also examine the effect of the imbalance degree of the trend direction ratio. We compare the simplest form of standard trend-following rules: time-series momentum (TMOM) and single moving average cross over (MACO), and the ordinal support vector machine (OSVM), a simple machine-classification-based non-trend-following rule. To minimize the risk of data snooping, we analyze data from major daily stock indices (the NYSE, NASDAQ, FTESEALL, and TOPIX) for the sample period of 1973 to 2019. The out-of-sample test period is from 1995 to 2019, when researchers began to report unstable and reduced profitability, especially in the U.S. markets.

Second, we introduce a machine-classification-based trend-following rule, the long-term trend direction classifier (LTDC). We adopt this approach to contrast the features specific to trend-following rules and examine the effectiveness of incorporating simple machine-classification-based remedies in varying market conditions involving the problem of non-stationarity

and the class imbalance. With only the target price series, the LTDC learns the *stable* longterm MA-smoothed trend direction via a support vector machine (SVM) classifier in relation to the trend reversal signs *sensitively* detected via short-term daily return moments. The difference between OSVM and LTDC is the target to learn: OSVM learns the *raw* short-term (e.g., monthly) stock price change direction, but the LTDC learns the MA *smoothed* long-term stock price trend direction that realizes stable learning. We set the other conditions the same: Both OSVM and LTDC employ SVM which is advantageous for varying market conditions compared to many classifiers, and contain remedies for an imbalanced direction class.

This study offers two contributions to the literature. First, the results empirically illustrate the underlying causes of the recent unstable performance, which, as the study demonstrates, does not imply the loss of profitability. In addition to the significant advantage of trend-following rules in avoiding downside risks, we find a discrepancy in the optimum averaging window size between positive and negative trend phases, which is widened by higher positive trend direction ratio, reflecting the degree of discrepancy. A higher positive trend direction ratio leads to poorer performance relative to buy-and-hold returns (BHR), and the effect exceeds that of volatility. That is, the discrepancy creates a dilemma regarding which trend direction to emphasize when selecting the main control parameter, the averaging window, resulting in the unstable performance of trend-following due to inappropriate averaging window size to adapt to the market conditions black.

Second, we provide effective suggestions to mitigate the dilemma of averaging window selection by focusing on the essential difficulty of trend-following rules. We demonstrate the effectiveness of incorporating the simplest possible machine classification-based remedies into standard trend-following rules to alleviate the negative effects of non-stationarity with class imbalance on performance. We validate the advantages of trend-following rules in protecting against long-term trend reversals via bootstrap significance tests. Among the compared rules, the LTDC machine-classification-based trend-following rule has the highest returns and robustness most of the time. Particularly, we extend the advantage of downside protection of trend-following to markets with higher positive direction ratios. Moreover, we suggest a simple data-based guideline to select the optimum averaging window: the mean return level of the target price series as a practical indicator reflecting both the key information of the trend direction ratio and the trend strength. This helps to attain profit closer to the maximum achievable level.

Our results confirm that trend-following rules demonstrate their true value in markets with sufficient opportunities for trend reversals. Their protection ability relative to BHR is not fairly rewarded during a stable, upward trend period. This explains why the profitability of trend-following rules appears more commonly in commodity markets than in stock markets. Our results also suggest that the recent reduced profitability of trend-following rules relative to BHR in U.S. equity markets could be explained by the insufficient downside risks; that is, it does not outperform BHR due to the prolonged upward trend periods in the dataset. We extend the advantage of trend-following rules to markets with higher positive trend direction ratios by providing effective remedies to mitigate the dilemma of which trend direction phase to emphasize in averaging window selection. These remedies are commonly applicable for all trend-following rules with any machine classifier, enabling further enhancement of the downside risk protection ability of the trend-following rules.

The remainder of the paper is organized as follows. Section 2 summarizes the related studies. Section 3 explains the trading rules compared including the proposed LTDC method and remedies. Section 4 describes the data, while Section 5 outlines the research design. Section 6 presents the results of a comprehensive comparative out-of-sample experiment. Section 7 provides discussion on the remaining problems, and section 8 concludes.

2 Related studies

2.1 Trend-following studies in finance

Trend-following strategies are widely used among practitioners and are receiving growing scholarly attention because of their potential for high profitability and because they outperform BHR. See, for example [9–14]. Above all, trend-following strategies are as strong during market stress periods with downside risk or high volatility phases [13, 15, 16]. In particular, prior studies document the profitability of trend-following rules in volatile assets [17–19].

However, the recent reduced or unstable profitability of trend-following rules in equity markets has become a subject of discussion, casting doubt on their effectiveness. In U.S. equity markets, particularly for indices and large stocks, some studies report diminished or unstable profitability from trend-following rules since around 1990 [13, 20–22]. Others report a similar tendency in the equity markets of other developed countries [23] and foreign exchange markets [24], though no studies report evidence of a decrease in profitability in commodity futures markets [25].

Empirical research suggests several factors leading to diminished or unstable profitability from trend-following rules in U.S. equity markets. According to [26], the decrease in several anomalies, including the momentum effect, is attributed to increased liquidity and trading activity. The study about the out-of-sample and post-publication return predictability by [27] suggests data mining effects, as well as the possibility that traders learn about mispricing from academic publications.

Focusing on the fundamental problems of non-stationarity, a few studies provide theoretical evidence that the unstable profitability of trend-following rules is caused by the difficulty of keeping the trading model optimal, where the averaging window size and trend strength are the key factors. Assuming an autoregressive return process, [28] study the relationship between averaging window size and the trading performance of MA rules theoretically, and find that the dominant factor affecting trading performance changes from autocorrelation to the drift around a few months' look-back period. Moreover, their in-sample study demonstrates that the drift (autocorrelation) phase is characterized by an increase (decrease) in the Sharpe ratio as a function of the averaging window size.

Authors in [29] theoretically compare and contrast TMOM and MA rules to summarize important properties of trend-following rules. Assuming an autoregressive return process, they demonstrate a theoretical significant positive effect of trend strength on the similarity in forecasting accuracy between the trading rules of TMOM and MA. Given the difficulty of identifying trend strength and the time-varying optimal averaging window size, they conclude that maintaining the optimum averaging window is difficult, and that the robustness to the changes in the averaging window size determines whether MA outperforms TMOM in terms of profitability. However, the overall picture of how trend strength and averaging window affect the profitability behavior of trend-following is not yet empirically clarified.

2.2 Measures for non-stationarity with class imbalance in machine learning

Whereas non-stationarity with class imbalance has become a critical issue in applying machine learning to real world problems, the combined problem is addressed by very little work [5, 7]. There are several measures developed to overcome non-stationarity without class imbalance problem, which can be classified into the following three categories: 1) adaptive base learners, 2) adaptive training set formation, and 3) ensemble techniques [4, 5]. The second category modifies the training set for better adaptation, which we can further decompose into windowing techniques and weighting techniques [5]. For the purpose of realizing practical

performance, however, [4–6] criticize the application of general-purpose theory-oriented remedies, and emphasize the importance of designing the simple adaptation algorithm customized to the specific nature of the non-stationarity in each application problem.

This study focuses on the windowing techniques in the second category because 1) it is the simplest and 2) the standard trend-following trading rules are classified as windowing method in that it modifies the target trend direction whose scale is controlled by averaging window or look back period. Windowing techniques select an appropriate set of previous data to train a classifier. How to use the past data streams depends on the type of non-stationarity the user confronts, the nature of available data, and required computational costs. The most important drawback of the windowing techniques is that it is impossible to determine the appropriate window size for any given problem in advance [5]. Moreover, the user confronts a stability-plasticity dilemma in optimizing window size between small window (fast adaptivity) and large window (better generalization) [30].

The remedies developed for class imbalance under stationarity are classified as: 1) resampling techniques, 2) algorithm-level methods, 3) ensemble learning. However, they are not effective for non-stationarity data streams because they cannot handle drifts that affect classification boundaries [7]. Employing evaluation measures giving more weight to the minority class is simple and easily applicable, and reported as effective to handle imbalanced data stream [31, 32]. Whereas there are several proposed measures such as G-mean, recall, and the Kappa statistic, for example, Kappa statistic is reported as advantageous relative to the others for real-time unbalanced class measure of accuracy [8, 33].

3 Trading rules and remedies

3.1 Standard trend-following rules

3.1.1 Time series momentum rule. Let *t* be the end date of the most recent month. The observed series of raw daily closing prices at date *t* are $P = \{P_1, P_2, ..., P_t\}$. The trend-following rules make trading decisions based on the direction of the price trend during the past d_L days up to date *t*, where d_L is the *look-back period* or *averaging window*. The difference between the MA and TMOM is how we obtain the price trend. For both rules, the time scale of the long-term trend changes as d_L changes.

TMOM introduced by [11] differs from the conventional cross-sectional relative returns in that it focuses purely on a security's own past returns. The TMOM rule generates a buy (sell) signal at date *t* if the past d_L -day return is positive (non-positive). That is, the TMOM generates a buy signal at date *t* if

$$TMOM_{t,d_L} = P_t - P_{t-d_L} > 0,$$
 (1)

and a sell signal when $TMOM_{t,d_L} \leq 0$. The source of the conventional cross-sectional momentum profits originates mostly from the time-series dependence in realized returns rather than the cross-sectional factors according to studies of momentum profit decomposition by [34, 35]. A 12-month look-back period and a one-month holding period is reported to generate the highest profitability for all securities [11].

3.1.2 Moving average-based rule. MA-based trading rules have many variations, and we use the most popular formulation, MACO. Using data on 28 commodity futures markets, [10] demonstrate the superior performance of dual MA crossovers and channel strategies relative to momentum strategies. They emphasize that employing the typical research design for momentum strategies using intermediate investment horizons is the key for a profitable result using MA strategies. In addition, they document that weakly significant individual results can be significantly powerful when they are aggregated across markets. We obtain the long-term

smoothed price at date t, y_t , by the d_L -day simple MA, formulated as

$$y_t = \frac{1}{d_L} \sum_{i=0}^{d_L - 1} P_{t-i}.$$
 (2)

Here, the averaging window, d_L , controls the degree of smoothing. Let z_t be the short-term MA with an averaging window of d_s , where $d_s < d_L$. The MACO generates a buy signal on date t when

$$MACO_{t,d_L,d_s} = z_t - y_t = \frac{1}{d_s} \sum_{i=0}^{d_s-1} P_{t-i} - \frac{1}{d_L} \sum_{i=0}^{d_L-1} P_{t-i} > 0,$$
(3)

and a sell signal when $MACO_{t,d_L,d_s} \leq 0$. In the simplest form, we use the price at time *t*, P_t , as the short-term MA, z_t .

3.2 Long-term trend direction classification rule

3.2.1 Overview of the method. LTDC is a machine-classification-based trend-following rule which is designed as simplest possible to realize effective handling of non-stationarity with class imbalance that investors encounter to avoid losses from large trend reversals. The LTDC trading rule predicts future long-term trend direction by stably learning the past relationship between the MA-smoothed long-term trend direction and sensitively capturing the varying market conditions expressed as the recent return moments using the SVM classifier.

Fig 1 shows the overall architecture of the LTDC system. The system consists of two steps before the unseen data classification: feature representation and classification model (re-)construction. Given raw daily price time-series P and averaging window d_{I} , the feature representation step prepares the class labels of y (d_L -day MA of P). In the next step, Δy is computed as daily return time series of y during past d_m days, and the obtained moment statistics of Δy is used as input vectors **x**. Here, the feature is that class labels are generated not from the *raw* price series P but from the MA *smoothed* price series y representing the long-term trend movement. We directly monitor the changes in return density shape (i.e., market conditions) through the moment statistics of the short-term daily smoothed price change series. Any machine classifier can be used in our framework, but we select binary SVM classification for its robustness under concept drift. Parameter optimization is by grid search through k-fold CV using maximum training data period to keep information of rare large trend reversals, which reduces the need of frequent model re-construction. We select the model based on Kappa statistics as they provide the sensitivity to detect a minor downward trend direction. Provided the input vectors x (recent return moments describing market condition), SVM classifies the corresponding direction of target trend y. LTDC generates a buy (sell) signal if the classified direction is positive (negative).

We provide a detailed explanation of the respective parts and the implementation below.

3.2.2 Support vector machine classification. The SVM was developed by [36] and it has since been widely applied because of its excellent classification performance. Among various methods, we select SVM for machine classifier of LTDC because it is simple but realizes robust online learning under non-stationarity [4, 30]. Actually, the SVM is one of the most used classifiers in online concept drift learning studies [37]. In forecasting stock price direction, SVM is one of the best algorithms [38–41]. Specifically, it is reported as advantages in environments with non-stationarity and noise relative to competing methods [39, 42].

The SVM is based on the principle of structural risk minimization and statistical learning theory. The basic idea behind SVM is to map the original data points in an input space to a



Fig 1. Schematic layout of the long-term trend direction classification method.

high-dimensional feature space using linear classification. The learned classification rule is then mapped back to the original input space. The standard model, C-support vector classification, is as follows. Consider the training dataset **x** composed of N input vectors $\mathbf{x}_1, \dots, \mathbf{x}_N$ corresponding to target labels y_1, \dots, y_N , where $y \in \{-1, +1\}$. We set labels y_i to -1 (+1) if the trend direction is downward (upward). Let $\phi(\mathbf{x})$ denote a fixed feature space transformation for some weight vector **w** and bias parameter *b*. Then, the binary soft margin classification problem is

$$\underset{\mathbf{w},b,\xi}{\operatorname{arg\,min}} \quad \frac{1}{2} ||\mathbf{w}||^2 + \frac{C}{N} \sum_{n=1}^N \xi_i, \ (C > 0), \ \text{subject to} \ \ y_i(\mathbf{w}^T \phi(\mathbf{x}) + b) \ge 1 - \xi_i, \tag{4}$$

where $\xi_i \ge 0$, $i = 1, \dots, N$ is the slack variable tolerating misclassification and parameter *C* controls the trade-off between a large margin and misclassification errors. That is, a larger *C* increases the rate of recognition but reduces generalization performance through overfitting.

3.2.3 The stable class label of the long-term trend direction. The LTDC target to classify is not the ordinal *raw* (typically daily) price change direction but the *smoothed* long-term price trend direction, which facilitates the learning task of distinguishing the large downward trend reversals from noisy fluctuations. Among the various methods for modifying the training set for better adaptation, the windowing technique is the most common approach and provides clearer probability estimation on unseen data [4, 5]. To obtain the long-term trend, we select MA smoothing as its performance is superior to that of the TMOM rule [13, 29].

Fig 2 illustrates the algorithm that assigns the trend direction class label. The gray line illustrates the raw price series, *P*. We calculate the d_L -day simple MA-smoothed price up to time *t*, y_t , as in Eq.(2), and plot $y = \{y_1, y_2, \dots, y_t\}$ as the thick red and blue lines. We obtain the green line, \tilde{y} , by further smoothing *y* to locate the local minima and maxima of *y*. We further smooth the line by applying the d_f -day MA extended toward both sides via a Nadaraya-Watson kernel regression [43] with a box kernel of bandwidth d_f . Along the smoothed trend line, *y*, the



Fig 2. Algorithm to assign the smoothed price trend direction labels. Local maxima (minima) between adjacent intersections of the smoothed price trend (thick red and blue lines) and further smoothed price trend (green line) are set as the trend direction reversal points of the smoothed price.

https://doi.org/10.1371/journal.pone.0276322.g002

highest (lowest) point in orange (light blue) between the adjacent green intersections is the local maximum (minimum). Then, we assign a section from the local minimum (maximum) to the local maximum (minimum) as an "Up" ("Down") direction class, as colored in red (blue).

The class assignment algorithm above is inspired by the MA-based technical indicators, the golden cross and the death cross, but we modify them such that the buy/sell signal timing is on the trend reversal point of the smoothed price to learn the market status by the trend direction class. We control the target scale of trend reversals by the averaging window, d_L , which ranges from several days to several years.

3.2.4 Input vectors to sensitively capture market condition changes. Given the smoothed price at time t, y_t , as in Eq (2), we formulate the daily price change at time t as $\Delta y_t = \log y_t - \log y_{t-1}$. Let x at time step t be the smoothed price change series during d_m days such that $x(t, d_m) = \{\Delta y_{t-d_m+1}, \dots, \Delta y_t\}$. We use the recent moment statistics of $x(t, d_m)$ as input vectors, where the statistics are centralized about the mean for orders larger than one. The optimum order of the moments to use depends on the nature of the target movement and the available information. We focus on the orders up to four, the mean, variance, skewness, and kurtosis to avoid overfitting and to simplify the interpretation of the results.

The advantages of using the moment statistics of the target series as input vectors are twofold. First, it requires no external data and model assumptions, avoiding the risk of model-misspecification due to market condition changes. Using the minimum number of input vectors is desirable to create the simplest model possible to mitigate the minority class data insufficiency problem. However, selecting the appropriate input vectors is a difficult task requiring expert knowledge of the target's behavior. Moreover, the best set of input vectors can change over time, which requires frequent updates. In our approach, instead of renewing the optimum set of input vectors to adjust for the changes in market conditions, we capture the market status change itself directly by the changes in the moment statistic.

Second, moment statistics summarize certain characteristics of the probability density shape patterns and thus provide an effective warning of trend reversals by directly capturing the non-stationary market condition. In particular, the third moment, skewness, indicates the degree of trend direction class imbalance. Various streams of research provide evidence of the effective warning ability of moment statistics. The literature on anomalies in prices reports evidence that variance and skewness contain information on future return movements and offer predictability (see, e.g., [44]). Further, several research streams report the effectiveness of moment statistics as completely data-driven signals of the approach of critical transitions, such as ecology, neuroscience, and medicine. Several more detailed reviews are provided by [45–47].

3.2.5 Long memory and remedies for class imbalance. Validation system is the k-fold blocked cross-validation of [48], because it is advantageous in time-series direction forecasting with small sample data compared to other procedures such as the standard out-of-sample procedure [49]. Parameter optimization is by a grid search.

All available data are partitioned into in-set samples used for training and out-of-sample used only to evaluate performance. In-set samples are partitioned into k sections of equal size. In the in-set training process, one block is withheld, as the in-set testing block, and the remaining k - 1 blocks are used for learning. Each of the k in-set blocks can be the in-set test block in the in-set training phase in turn. The directional accuracy of the candidate models is measured k times, and the average value of the k accuracy results is compared with candidate models to select the best model specification. The selected best model with the optimum parameter set is retrained using all k in-set training blocks to forecast the trend direction of the out-of-sample test block, which is used to evaluate trading performance.

In online learning environment, the training data period expands every time SVM is reconstructed, adding new data to use the full length, and partitioned into *k* sections of equal longer size. Adaptive block size is desirable to adjust to the drift length, and an appropriate fixed block size is a compromise between fast adaptivity (small size) and good generalization (large size) [30]. However, the size or variability of the respective block depends on the applications in which it is used [50]. For our research aim, loss aversion from large long-term trend reversals which rarely happens, it is critically important to hold long memory by holding longest available training data, which is also suggested by our preliminary test results.

The choice of model evaluation measure is especially important for avoiding trend reversal risks, particularly when the target data have concept drift with class imbalance. In stock markets, a high degree of class imbalance occurs in periods of stable upward trend phases. From the risk management perspective, risky events to avoid occur only rarely, but the correct detection of minor risky classes is more important than for non-risky classes. However, in the stock price trend direction classification problem, standard measures such as accuracy, evaluate the classification performance giving more weight to the relatively frequent, non-risky positive direction. Moreover, investors and traders tend to prioritize avoiding losses over seeking the same size gains [51]. Hence, we use an evaluation measure that places greater weight on the minority downward trend direction class for maximum exploitation of information about the downward trend class. Among these measures, we employ kappa statistics, which is one of the best evaluation measures receiving increased attention, particularly in the literature on online learning with concept drift and class imbalance [31, 32].

The specification of the directional evaluation measure, the kappa statistics, is as follows. Denote the true positive, false positive, true negative, and false negative as TP, FP, TN, and FN, respectively. Then, TP (FP) implies a correct (wrong) upward direction forecast; that is, a correct (wrong) buy decision. TN (FN) implies a correct (wrong) downward direction forecast; that is, a correct (wrong) sell decision. Accuracy is defined as

$$Accuracy = p_0 = \frac{TP + TN}{TP + TN + FP + FN}.$$
(5)

Then, Cohen's kappa statistic is

$$Kappa = \frac{p_{o} - p_{e}}{1 - p_{e}}, \ p_{e} = \frac{(TP + FN)(TP + FP) + (FP + TN)(FN + TN)}{(TP + TN + FP + FN)^{2}},$$
(6)

where p_o is the relative observed agreement identical to accuracy, and p_e is the expected agreement obtained when the same number of examples are observed in each class.

3.3 Standard machine-classification-based non-trend-following rule

We add a standard machine-classification-based non-trend-following rule, OSVM, to our comparison to contrast the differences between trend-following rules and non-trend-following rules. OSVM learns the direction of non-smoothed raw daily price change series ΔP following common manner in machine learning studies. The design of OSVM is formulated as identical with that of LTDC except that it classifies direction of *raw* price change series ΔP instead of that of *smoothed* price change series Δy . Corresponding to the replacement of target direction series from Δy to ΔP , the input vectors are also replaced from the moment statistics of Δy during past d_m days to those of ΔP during past d_m days.

4 Data

4.1 Data description

Selecting the dataset is important for an appropriate performance evaluation of a trend-following trading strategy, and [52] warn of the danger of data-snooping bias and recommend using only major stock market indices with sufficient length to enable the dataset to include longterm trend reversal information. This study uses the daily closing prices of four major daily stock market indices: the NYSE, NASDAQ, TOPIX, and FTSEALL. The selected indices are the weighted averages of all (or a major part of) the listed companies in each market. These are considered the most representative stock indices of the major countries. We collected these data from Datastream database for the period from 1973 to 2019. For all four stock indices, the in-sample trading period starts from January 1975, and the corresponding in-sample input data period starts from around the beginning of May 1973, which is the maximum averaging window 400 trading days before January 1975. The out-of-sample test period is from January 1995 to December 2019. This period covers the lower, unstable profitability of trend-following rules in the U.S. equity market, especially for large stock and market indices [13, 20–22].

Fig 3 shows the movement of the log price and the corresponding daily returns for our full sample period. All four stock index price series have repeated upward and downward trend formations at the micro- and macro-scales. In the return sequences, we see periodic volatility clustering. Table 1 summarizes the basic statistics of the daily raw data and moment statistics, in addition to monthly directions. From Fig 3, we see that the long-term trend behavior of the TOPIX is quite different from that of the other indices. If we consider the long-term linear trend line, after the 1990s, only the TOPIX exhibits a sideways trend, while the other indices





Table 1. Descriptive statistics.

Stock index		Monthly direction					
	Sample size	Mean	Variance	Skewness	Kurtosis	Up (%)	Down (%)
NYSE	11769	0.00026	0.00011	-1.055	27.364	334 (59.7)	225 (40.3)
NASDAQ	11769	0.00037	0.00015	-0.306	12.327	334 (59.7)	225 (40.3)
FTSEALL	11794	0.00026	0.00011	-0.272	11.505	340 (60.8)	219 (39.2)
TOPIX	11496	0.00014	0.00013	-0.378	13.414	314 (56.1)	246 (43.9)

Full sample period from 1973 to 2019.

https://doi.org/10.1371/journal.pone.0276322.t001

show a globally upward trend. The monthly direction ratio of the TOPIX in <u>Table 1</u> also has the highest downward ratio among the four indices, but the difference from the other indices is not that large.

4.2 Trend direction ratio imbalance problem

We must note the serious effect of changing the target trend scale in the difficulty of addressing the trend direction identification problem. The price trend becomes longer-term as the averaging window, d_L , increases to smooth the trend further, resulting in a more serious trend direction ratio imbalance. A large positive trend direction ratio leads to deficient information about downward trend direction, rendering the trend direction identification problem difficult to address.

Using the full sample period data from 1973 to 2019, Fig 4 illustrates the increases in the positive trend direction ratio as the target trend scale to measure directional (i.e., return frequency) changes, from daily to two-yearly, for all stock market indices. For the U.S. stock market indices, the positive direction ratio increases to over 80% at the two-yearly frequency, while it is at a level slightly above 50% at the one-day change direction. In particular, it is difficult to predict the trend reversal of large d_L after a prolonged period of continued upward direction because information on the downward direction class in the past data, used to construct the model for trading decisions, is insufficient. Among the four indices, the degree of positive/negative direction class imbalance is the smallest for the TOPIX, and the difference with other indices increases on longer time scales.

5 Research design

We next discuss the design and setting of the out-of-sample back test to compare four trading rules: TMOM, MACO, LTDC, and OSVM.

5.1 Computational set-ups

The TMOM and MACO rules are given in Eqs (1) and (3) with $z_t = P_t$, respectively. We described the LTDC in Section 3.2 and OSVM in Section 3.3, respectively. We use the *ksmooth* function in R with bandwidth d_f to obtain the further smoothed price trend of LTDC, \tilde{y} , which we require for classifying the long-term trend direction. It is desirable that d_f changes as d_L changes, and we set $d_f = 2d_L$.

As for the SVM classification of OSVM and LTDC, the computational details are as follows. We choose the short-term memory parameter for obtaining the moment statistics from $d_m = \{5, 10, 15, 20\}$ days. We implement the SVM using the kernlab [53] package interface for libsvm [54], which includes an efficient version of the sequential minimization-optimization algorithm. For the nonlinear mapping ϕ from the input space to the high-dimensional feature space, we use the Gaussian radial basis function (RBF) kernel, which is the most suitable for our purpose and is widely applied to unknown general time series data. The soft margin penalty *C* controls the trade-off between the size of the margin and hyperplane violations. The RBF kernel requires parameter γ , which controls the influential range of a single training example. We set $C = \{2^{-3}, \ldots, 2^3\}$ and $\gamma = \{2^{-3}, \ldots, 2^3\}$. The number of in-set blocks *k* is set as four.

5.2 Experimental scheme

The four trading rules make buy/sell trade decisions given the past daily price data. We evaluate the one-month ahead out-of-sample trading performance. We consider a monthly trading



Time scale for change rate (Log scale)

Fig 4. Changes in the positive direction ratio as the time scale of the change direction increases.

https://doi.org/10.1371/journal.pone.0276322.g004

decision, as a one-month holding period achieved the best trading performance in a study by [11]. We evaluate the results using a prequential approach, which is desirable in a streaming setting with evolving data [7, 33]. Considering the loss aversion from long-term trend reversals, such as a stock market crash, we use an expanding (landmark) window that enables us to exploit the maximum information in the data.

The start of the in-sample trade performance evaluation period is 1975.1, and the input data period starts from 400 trading days prior to obtain Eqs (1) and (2) with the maximum averaging window. The sliding out-of-sample test period starts from 1995.1 and ends in 2019.12, during which previous studies report an insignificant profitability of trend-following rules. A buy/sell decision for the trade at 1995.1 is made after 1994.12 based on the models using in-sample training data up to 1994.12. We evaluate the out-of-sample one-month returns from 1994.12 to 1995.1. The buy/sell decision of the second test month, from 1995.1 to

1995.2, is made based on the in-sample data up to 1995.1, and the same applies to the following months.

For the SVM classification of OSVM and LTDC, the model is reconstructed every five years, which is determined as a compromise between computational cost and model durability under the constraint of available data. The construction of the first model is based on the data up to 1994.12 and the first (second) monthly trade decision for 1995.1 (1995.2) is made when the input vector up to 1994.12 (1995.1) becomes available. The application of the first model continues until the monthly trade decision for 1999.12. The second model is constructed using data up to 1999.12, and the first (second) monthly decision for trade from 1999.12 (2000.1) to 2000.1 (2000.2) occurs in 1999.12 (2000.1). This process continues until the last test month, 2019.12, where we apply the fifth model using data up to 2014.12 and the input vector up to 2019.11 for the prediction.

The subsequent section compares the out-of-sample test results of the four trading rules from 1995.1 to 2019.12 to analyze the causes of the unstable profitability of trend-following rules and explore the potential remedies.

6 Results

6.1 Unconditional performance analysis

6.1.1 Profitability changes by target index and averaging window. The out-of-sample comparative study shows that changing the target index and averaging window, d_L , has a significant effect on unconditional trading performance, and we see common patterns for the trend-following rules. Fig 5 shows the cumulative monthly returns of the trend-following strategies by trading four major stock market indices in the recent period with different d_L lengths. We find that three trend-following rules, MACO, TMOM, and LTDC, mostly outperform BHR for the TOPIX, but their performance for the other indices depends on d_L . Moreover, the excess return of the BHR of trend-following rules is higher for lower BHR indices, and the optimum d_L for each index is mostly common among the trend-following rules. Thus, the effect of changing the trading rule is sometimes outweighed by the change in the target index or d_L . Among the four rules, the performance of LTDC is best mostly around the common optimum d_L , with no negative returns for all indices and d_L .

Selecting the averaging window appropriate for each market property is quite effective for trend-following rules to significantly outperform BHR by avoiding losses from trend reversals. Fig 6 illustrates the evolution of the cumulative returns of trend-following strategies with





https://doi.org/10.1371/journal.pone.0276322.g005



Fig 6. Evolution of cumulative monthly returns at the common optimum averaging window for all trend-following rules. The selected common optimum averaging window is d_L = 400 for NASDAQ, 200 for the NYSE and FTSEALL, and 100 for the TOPIX.

respect to the common optimum d_L , wherein the average of all three trend-following rules' returns are maximized: $d_L = 400$ for NASDAQ, 200 for NYSE and FTSEALL, and 100 for TOPIX. We find that the machine-classification-based non-trend-following rule, OSVM, performs poorly in most cases, significantly underperforming the trend-following rules, especially for the TOPIX. The performance differences between the trend-following rules and BHR widens notably during a large price decline, implying the common advantage of trend-following rules in avoiding losses from trend reversals. As a result of the significant effect of market trend conditions common to trend-following rules, we observe the coincidence of the trading decision of LTDC and TMOM for the NASDAQ with $d_L = 400$. In our experiment, this coincidence occurs only with the combination of the NASDAQ with $d_L = 400$, which has the simplest classification problem. We presume that the lowest noise level of the 400-day MA direction makes the LTDC learning (by matching return moments) degenerate to focus only on the past price mean movement. The LTDC has the best performance of all methods.

The features illustrated in Figs 5 and 6 are in line with results reported by [13] and [29] in that the TMOM and MA-based rules have many similarities in their trading behavior. However, their finding that the MA-based rules outperform the TMOM rule is not clearly supported: we observe that the selection of d_L and the market conditions affect the relative performance of the TMOM and MACO. Alternatively, we find significantly different behavior between the three trend-following rules and the non-trend-following rule. This result suggests that the similarity and the nature common to trend-following rules stem from the focus on the direction of the *long-term price trend change* instead of the *monthly raw price change*, as it is the only difference between the OSVM and LTDC.

6.1.2 Downside risk protection. Our results validate the significant ability to protect against downside risks rather than high volatility as a property common to trend-following rules. We evaluate the risk tolerance of each rule by the standard risk measures: Sharpe ratio, Sortino ratio, and maximum drawdown. We set the risk-free rate at zero. Table 2 summarizes the risk tolerance statistics of the out-of-sample test period with reality-check significance by [55], corresponding to the hypothesis that the performance measures outperform those of

Stock index	d_L	Sharpe Ratio ⁽¹⁾				Sortino Ratio ⁽²⁾				Maximum drawdown ⁽³⁾						
		LTDC	MACO	ТМОМ	OSVM	BHR	LTDC	MACO	ТМОМ	OSVM	BHR	LTDC	MACO	тмом	OSVM	BHR
NYSE 10 20 50 100 200 300 400	10	0.219	0.035	0.135	0.420	0.534	0.312	0.050	0.204	0.621 0.767	0.767	0.667	0.549	0.637	0.376** 0	0.552
	20	0.187	0.037	0.084			0.260	0.053	0.123			0.483***	0.633	0.524		
	50	0.447	0.168	0.349			0.672	0.249	0.540			0.465	0.497	0.323***		
	100	0.396	0.371	0.471			0.597	0.574	0.723			0.578	0.316***	0.472		
	200	0.604	0.524	0.497			0.926	0.795	0.745			0.375**	0.449	0.441*		
	300	0.427	0.367	0.420			0.648	0.538	0.626			0.429*	0.569	0.514		
	400	0.611	0.345	0.321			1.001	0.506	0.476			0.293***	0.466*	0.585		
NASDAQ	10	0.405	-0.056	0.150	0.426	0.560	0.606	-0.079	0.216	0.620	0.836	0.757	0.853	0.768	0.667**	0.750
	20	0.481	0.097	0.192			0.711	0.140	0.287			0.750	0.705*	0.602**		
50 100 200 300 400	50	0.479	0.334	0.246			0.739	0.502	0.369			0.493***	0.502***	0.559***		
	100	0.476	0.417	0.213			0.719	0.649	0.311			0.460***	0.441***	0.706		
	200	0.443	0.427	0.386			0.674	0.656	0.598			0.459***	0.558***	0.458***		
	300	0.589	0.420	0.510			0.937	0.657	0.791			0.538***	0.550***	0.543***		
	400	0.560	0.518	0.560			0.859	0.822	0.859			0.564***	0.393***	0.564***		
FTSEALL	10	0.276	-0.017	-0.083	0.215	0.377	0.400	-0.024	-0.116	0.294	0.527	0.345**	0.504	0.618	0.463	0.469
	20	0.250	-0.179	-0.120			0.346	-0.248	-0.166			0.462	0.634	0.656		
5 10 20 30	50	0.167	-0.118	0.274			0.245	-0.164	0.425			0.450	0.617	0.337*		
	100	0.417	0.098	0.377			0.635	0.146	0.593			0.333*	0.503	0.341*		
	200	0.549	0.502	0.371			0.867	0.801	0.565			0.237***	0.311*	0.340*		
	300	0.439	0.352	0.317			0.690	0.534	0.487			0.319*	0.443	0.495		
	400	0.308	0.306	0.226			0.470	0.462	0.338			0.395	0.466	0.571		
TOPIX	10	0.317	0.027	0.478*	0.050	0.111	0.465	0.038	0.757*	0.069	0.155	0.492	0.599	0.334**	0.599 (0.595
2	20	0.143	0.223	0.435			0.212	0.333	0.695			0.475	0.421*	0.394		
	50	0.416	0.415	0.471*			0.648	0.641	0.741*			0.336**	0.352*	0.433		
	100	0.555*	0.509*	0.527*			0.885*	0.797*	0.840*			0.404	0.348*	0.434		
	200	0.373	0.350	0.325			0.585	0.528	0.501			0.492	0.629	0.554		
	300	0.325	0.420	0.328			0.498	0.659	0.503			0.460	0.359*	0.460		
	400	0.136	0.516	0.089			0.199	0.820	0.128			0.565	0.387	0.584		

Table 2. Risk tolerance statistics.

(1) and (2) are annualized. (3) is the maximum drawdown from the peak. The reality check corresponds to the hypothesis that the performance measures outperform those of the BHR. Statistical significance at the 1%, 5%, and 10% levels are indicated by ***, **, and *, respectively.

https://doi.org/10.1371/journal.pone.0276322.t002

BHR, where we use the stationary block bootstrap method of [56] with the optimum block length determination by [57]. High volatility significantly affects the unstable performance of trend-following rules [13, 17, 18]. However, Table 2 shows that the performance of trend-following rules relative to the BHR is best for maximum drawdown, and their Sortino ratios are better than their Sharpe ratios. Most trend-following results for the lowest BHR market, TOPIX, are significant, indicating that trend-following rules are advantageous for such stagnating (i.e., not stably upward) market conditions. In addition, the d_L minimizing maximum drawdown is the same or smaller than the return-maximizing d_L . For all risk indicators, the LTDC has the best overall performance, especially at around the optimum d_L , where it outperforms the BHR, suggesting the effectiveness of incorporating machine classification-based solutions to avoid downside risks. OSVM offers significant drawdown risk protection for the NYSE and NASDAQ, whose cause we discuss in more detail later.

The stability of downside risk aversion of the LTDC is the best among the four rules from Figs 7 and 8, implying the effectiveness of trend-following with remedies for class imbalance





https://doi.org/10.1371/journal.pone.0276322.g007



and non-stationarity to avoid downside risks. Fig 7 depicts the time series plot of the drawdown for each rule since the start of the test period in 1995.1 at the d_L that maximizes the average returns of the three trend-following rules, as in Fig 6. Compared to the BHR (green line), the drawdown size of the LTDC (bold black line) is significantly smaller most of the time, exhibiting a stable drawdown protection ability. This is in contrast to the other rules, which experience periods with significantly underperforming BHR for some indices. Fig 8 compares the quantiles of the annual negative returns at the d_L that maximizes the average returns of the three trend-following rules. In Fig 8, the trend-following rules are mostly above the BHR, while the OSVM (blue line) is mostly overlapping the BHR, except for the NYSE and NAS-DAQ, whose positive direction ratio is the first and second among the four indices, respectively. Most of the time, the LTDC has the smallest loss, especially in terms of the largest downside risk.

6.1.3 Performance robustness. LTDC is robust in the mis-selection of the averaging window size and realizes an enhanced adaptation to market condition changes. Authors in [29] emphasize the difficulty of selecting the time-varying return-maximizing averaging window, and show that the forecast accuracy of trend-following rules depends on the robustness to changes in d_L . They propose a measure of performance robustness based on the average Sharpe ratio over all averaging window sizes. We follow their approach to obtain an indicator

for robustness to the change in the averaging window size, expanding the set of target risk measures.

Table 3 reports the Sharpe ratio, Sortino ratio, and maximum drawdown averaged over all d_L values of $d_L = 10, 20, 50, 100, 200, 300, and 400$. We test the equality of the d_L average of the risk statistics for each pair of rules. Table 3 also presents the reality check p-values following [55] based on the stationary block bootstrap by [56, 57]. Table 3 shows that the LTDC method has the best overall robustness to the change in d_L , with only minor performance fluctuations among the target indices. With respect to the Sharpe and Sortino ratios, the LTDC outperforms the rules for all indices except the TOPIX. Regarding maximum drawdown, the performance robustness of the trend-following rules is mostly stable for all target indices, in contrast to the OSVM, which differs considerably for each index.

The LTDC has a significant advantage in terms of robustness for the FTSEALL. Trendfollowing rules are commonly advantageous for the TOPIX. On the other hand, the OSVM tends to be advantageous for the NYSE and NASDAQ, except for the maximum drawdown of NASDAQ. Fig 4 shows the positive trend direction ratios for the NYSE, NASDAQ, FTSEALL, and TOPIX, in that order. Hence, the relative performance patterns are consistent with our conjecture that high positive trend direction ratios exacerbate the performance of trend-following rules, which we subsequently validate. The validity of this conjecture implies that the outperformance of the LTDC over the MACO and TMOM for FTSEALL suggests a significant improvement in dealing with the class imbalance problem by combining machine classification with trend-following rules to improve the adaptation of the rule to varying market conditions.

6.2 The mechanism of performance instability

We focus on the trend direction phase as a significant factor in the unstable profitability of trend-following rules. Among the potential market influences, [29] theoretically demonstrate the significant effect of trend strength and the averaging window, assuming an autoregressive return process. Moreover, they emphasize the importance of empirically estimating the trend strength for improving the trading performance under uncertainty. We consider the BHR during a calendar year as a simple measure of trend direction and strength, computable solely from price series.

6.2.1 Effect of trend direction phase on trade decision accuracy. The difficulty of the buy/sell decision changes according to the trend *direction* phase, and the most difficult phase differs for trend-following and non-trend-following rules. Fig 9 shows the kernel density estimates of the trade decision accuracy of each trading rule classified by the three trend direction phases of the target price index: positive, negative, and trendless. Using 25 years of trading decisions in the out-of-sample test results, the years with higher (lower) annual market returns (BHR) than 0.1(-0.1) are classified as a positive (negative) trend phase, and those within ± 0.1 are classified as a trendless phase. We compute the accuracy of the trading decisions in a calendar year using Eq (5) for each d_L and index, evaluating the correctness of 12 monthly trading decisions given the target actual raw monthly price change directions. In a strict sense, evaluating the accuracy of predicting the direction of *raw* price change for the non-trend-following rule, but for the three trend-following rules, it is that of the *smoothed* trend. We take the average accuracy of the trend-following rules across all d_L values.

In Fig 9, the prediction accuracy in the positive trend direction phase (red) is located at the extreme right for all trading rules, implying the highest accuracy phase, or the easiest phase for which to predict the direction. However, the most difficult phase to predict the monthly trend

Stock index		Trading rule						
		LTDC	MACO	ТМОМ	OSVM			
	Av	erage Sharpe ratio ov	ver all d_L					
NYSE	Average Sharpe ratio	0.413	0.264	0.325	0.420			
	P-value	LTDC	0.232	0.410	0.978			
		MACO	NA	0.320	0.498			
		ТМОМ	NA	NA	0.668			
NASDAQ	Average Sharpe ratio	0.491	0.308	0.322	0.426			
	P-value	LTDC	0.154	0.102	0.664			
		MACO	NA	0.840	0.610			
		ТМОМ	NA	NA	0.636			
FTSEALL	Average Sharpe ratio	0.344	0.135	0.195	0.215			
	P-value	LTDC	0.072*	0.124	0.592			
		MACO	NA	0.278	0.756			
		ТМОМ	NA	NA	0.916			
TOPIX	Average Sharpe ratio	0.324	0.351	0.379	0.050			
	P-value	LTDC	0.700	0.312	0.220			
		MACO	NA	0.628	0.202			
		ТМОМ	NA	NA	0.170			
-	Av	erage Sortino ratio or	ver all d_L	·				
NYSE	Average Sortino ratio	0.631	0.395	0.491	0.621			
	P-value	LTDC	0.224	0.408	0.996			
		MACO	NA	0.358	0.540			
		ТМОМ	NA	NA	0.706			
NASDAQ	Average Sortino ratio	0.749	0.478	0.490	0.620			
	P-value	LTDC	0.228	0.160	0.610			
		MACO	NA	0.932	0.712			
		ТМОМ	NA	NA	0.688			
FTSEALL	Average Sortino ratio	0.522	0.215	0.304	0.294			
	P-value	LTDC	0.086*	0.158	0.546			
		MACO	NA	0.296	0.832			
		ТМОМ	NA	NA	0.976			
TOPIX	Average Sortino ratio	0.499	0.545	0.595	0.069			
	P-value	LTDC	0.680	0.302	0.216			
		MACO	NA	0.612	0.200			
		ТМОМ	NA	NA	0.176			
-	Average	maximum drawdow	vns over all d_L					
NYSE	Average maximum drawdowns	0.470	0.497	0.499	0.376			
	P-value	LTDC	0.978	0.676	0.114			
		MACO	NA	0.522	0.228			
		ТМОМ	NA	NA	0.122			
NASDAQ	Average maximum drawdowns	0.574	0.572	0.600	0.667			
	P-value	LTDC	0.350	0.906	0.144			
		MACO	NA	0.146	0.116			
		ТМОМ	NA	NA	0.232			

Table 3. Average risk tolerance statistics and the tests of equality comparison between two rules.

(Continued)

Stock index			Trading rule						
		LTDC	МАСО	ТМОМ	OSVM				
FTSEALL	Average maximum drawdowns	0.363	0.497	0.480	0.463				
	P-value	LTDC	0.020**	0.006***	0.350				
		MACO	NA	0.968	0.784				
		ТМОМ	NA	NA	0.822				
TOPIX	Average maximum drawdowns	0.460	0.442	0.456	0.599				
	P-value	LTDC	0.796	0.612	0.340				
		MACO	NA	0.504	0.294				
		ТМОМ	NA	NA	0.472				

Table 3. (Continued)

The hypotheses that the Sharpe and Sortino ratios (maximum drawdown) are higher (lower) than those of the BHR is tested using a reality check. ***, **, and * indicate statistically significant reality check p-values at the 1%, 5%, and 10% levels, respectively.

https://doi.org/10.1371/journal.pone.0276322.t003

direction differs for the trend-following rules and non-trend-following OSVM. The phase located at the extreme left is the trendless phase (green) for the trend-following rules, TMOM, MACO, and LTDC, but it is the downward trend phase (blue) for the non-trend-following OSVM. The relative difficulty of the trendless phase and downward phase for the LTDC is lesser than that of the other trend-following rules, possibly because it combines the properties of OSVM and trend-following rules.

6.2.2 Discrepancies in the optimum averaging window between risk phases. While market stress, such as downside risks and high volatility, influence the performance of trend-following rules [13, 15, 29, 58], how those factors affect unstable profitability is not yet clear. Fig 10 comprehensively illustrates the effect of such market risk conditions on the relationship between excess returns relative to the BHR and averaging window. We measure the market risk condition by the market trend direction sign or the market volatility level.

In Fig 10(a), we split the annual excess returns of each rule by the trend direction of the target stock price index in a calendar year; that is, the sign of the yearly BHR. Similarly, we divide the annual excess returns of each rule into two phases at *s.d.* = *C* to plot Fig 10(b), wherein *C* is the median of the standard deviation of each stock index. The horizontal axis of Fig 10 is the averaging window size, d_L , which we interpret as the trend scale of the direction to predict. The bold solid lines indicate the unconditional excess returns, while the thin solid and dotted



Fig 9. Kernel density estimates of price direction prediction accuracy by trend direction phase. Trend direction is classified according to yearly BHRs with the threshold at ± 0.1 . The accuracy of trend-following rules is averaged over all d_L values.

https://doi.org/10.1371/journal.pone.0276322.g009



Fig 10. Changes in the relationship between annual excess returns and averaging window by market risk condition. (a)Risk phase by market trend direction; (b) Risk phase by market volatility level.

lines indicate the risky and non-risky phases, respectively. Black, light blue, pink, and blue denote the LTDC, MACO, TMOM, and OSVM, respectively. The leftmost panels of Fig 10(a) and 10(b) merge the results for the four indices on their respective right side, summarizing the effect of changing d_L on the profitability of trend-following rules by trend direction phase.

From Fig 10(a), we find discrepancies in the optimal averaging window of the trend-following rules between the positive and negative trend direction phases. As the leftmost merged panel shows, the averaging window affects the excess return differently for each trend direction phase, and the significant outperformance in the negative trend direction phase compensates for the under-performance in the positive trend direction phase. This result is in line with the reported advantages of trend-following rules during market stress periods. The relationship between d_L and excess returns in the negative trend phase is an inverted U-shaped curve, whereas that in the positive trend phase is rising to the right. Hence, the optimal d_L tends to be smaller than that for the positive trend direction phase. According to [10, 11], setting a longer investment horizon, a larger d_L in our context, is more profitable, but our result suggests that it is only valid during a period matching a positive trend direction phase.

The performance of each trading rule varies for each stock market index. This varying pattern could be explained by the degree of discrepancy by trend direction phase, where the optimal d_L in the positive trend direction phase is critical. The relationship between d_L and the

excess returns of trend-following rules in the positive market return (BHR) phase (dotted lines) is upward trending for the NYSE and NASDAQ, gently inverted V-shaped for the FTSEALL, and downward-trending for the TOPIX. The difference between the optimal d_L in each trend direction phase is the largest for the NYSE and NASDAQ, followed by the FTSEALL and the TOPIX. Hence, it becomes more difficult to stay near the optimal d_L at trend reversals for the NYSE and NASDAQ than for the TOPIX. That is, dealing with sudden trend reversals, which is the problem of non-stationarity, is difficult for trend-following rules in trading the NYSE and NASDAQ relative to trading the TOPIX.

We argue that the large discrepancy of the optimum d_L size by trend direction phase is due to the stable long upward trend period, which we can empirically measure by the positive trend direction ratio. With respect to the out-of-sample test period from 1995 to 2019, the average positive/negative return ratios of the yearly trend direction of the NYSE, NASDAQ, FTSEALL, and TOPIX are 76%, 76%, 68%, and 52%, respectively.

For the target index series with a largely positive trend direction, such as the NYSE and NASDAQ, trading decisions based on the past price movement are overwhelmed by the positive trend direction by frequency, which leads to a greater emphasis on the upward trend direction phase. This corresponds to the case when a long-term upward trend phase continues, and simply riding the upward trend without considering protection against downside risk can generate profit. Hence, a larger d_L becomes more profitable because it captures the long-term trend with less noise. However, adapting to trend reversals after a long continued positive trend is difficult, as traders are unprepared to detect the signs of a trend reversal. The situation is different for a balanced index such as the TOPIX, where the trend direction class imbalance problem is not severe. The downside risk of the TOPIX is larger than that of the other indices, even in a positive trend direction phase at around $d_L = 100$, is desirable. With less discrepancy in the optimal d_L , therefore, it is easier to maintain the optimal d_L for the TOPIX.

The relative advantage of the OSVM for the NYSE might be explained by the smaller imbalance in the target trend direction for trading decisions. The time scale of the target direction of the OSVM is smaller than that of trend-following rules. Fig 4 shows that the positive direction ratio of the NYSE and NASDAQ is more balanced for the shorter target time scale, which is one month for the OSVM but more than six months for the three trend-following rules. For the NYSE and NASDAQ, we conjecture that the benefit of reducing noise by targeting the smoothed trend direction is outweighed by the negative effects of the large class imbalance.

Table 4 reports the Pearson correlation and Spearman rank correlation test results for the risk factors and excess returns in the same year. These results confirm the adverse effect of high positive direction ratios on the excess returns of the timing strategies. We take the average excess returns of each trading rule in a year across all d_L values. For the first risk factor, the positive direction ratio, we use the monthly direction ratio in a year to ensure a sufficient sample size. We find a high significance of the positive direction ratio for all trading rules, while the OSVM has a lower significance with a smaller absolute coefficient than that for the trend-following rules. Given the statistically significant negative effect of the larger positive trend direction ratios on the performance of trend-following rules, the outperformance of the LTDC relative to the MACO and TMOM for the FTSEALL in the downside protection robustness result of Table 3 can be understood as the increased adaptability of trend-following rules to a higher degree of class imbalance, implying the effectiveness of machine classification-based remedies for class imbalance.

The findings from Fig 10(b) according to market volatility level are mostly like those of Fig 10(a) by market trend direction. However, the overall differences in performance by market, especially in a risky phase (upper side of the panels) are less obvious in Fig 10(b) than in Fig 10(a).

Risk factor	Method	Pearson co	rrelation	Spearman rank correlation		
		Coefficient	P-value	Coefficient	P-value	
Positive direction ratio ⁽¹⁾	LTDC	-0.655***	1.46e-13	-0.623***	4.55e-12	
	MACO	-0.637***	1.06e-12	-0.568***	6.84e-10	
	ТМОМ	-0.653***	1.74e-13	-0.626***	3.27e-12	
	OSVM	-0.440***	4.56e-06	-0.494***	1.80e-07	
Market volatility ⁽²⁾	LTDC	0.561***	1.29e-09	0.256**	0.010	
	MACO	0.563***	1.07e-09	0.224**	0.025	
	ТМОМ	0.553***	2.39e-09	0.301***	0.002	
	OSVM	0.198**	0.048	0.147	0.144	

Table 4. Pearson and Spearman rank correlations between risk factors and excess returns averaged over all d_L values.

⁽¹⁾ Positive monthly direction ratio of the stock index in a calendar year.

⁽²⁾ Standard deviation of daily index returns in a calendar year. Results for each index are merged.

***, **, and * indicate statistically significant p-values at the 1%, 5%, and 10% levels, respectively.

https://doi.org/10.1371/journal.pone.0276322.t004

Table 4 confirms the correlation between the volatility level of the index and the excess return of each rule, where the average daily standard deviation of each stock index in a year represents market volatility. However, the coefficients and significance are smaller than those of the positive direction ratio. This is consistent with the results in Tables 2 and 3, which indicate that the trend-following rules are advantageous in terms of downside risk rather than volatility risk. A large trend reversal tends to accompany a high volatility phase. While high volatility is a possible factor accounting for the unstable performance of trend-following rules, we find that trend direction and strength are likely to be more essential factors in the trading behavior of trend-following rules.

6.3 Towards optimization of the averaging window size

We see that trend direction change (non-stationarity) and the trend direction ratio (class imbalance) significantly affect the trading performance of trend-following rules. The profit-maximizing d_L should be smaller in a downward trend phase than in an upward trend phase to ensure the sensitivity to capture trend reversal signals. Given the difficulty of predicting sudden trend reversals, the key is to keep the averaging window near the optimal level that balances the trade-off between the trend and reversal to adapt to each trend direction phase.

Fig 11 shows that the d_L that achieves the maximum trading profit best balances the tradeoffs. The figure plots the 25-year out-of-sample results, where the horizontal (vertical) axis is the average annual returns when the yearly trend direction of the target index is negative (positive). The black upward-sloping line shows the 25-year average positive/negative ratio of yearly change direction of the target index, indicating the degree of the yearly trend direction class imbalance of the target index. The gray 45-degree lines indicate the even yearly trend direction ratio. The red, pink, and orange figures in each panel indicate the first, second, and third profit-maximizing d_L values, respectively. The blue star located at the upper right end of the black upward-sloping line indicates the direction that maximizes the profit while keeping the balance of the yearly trend direction ratio. It also indicates the overall strongest trend point along the trend direction balancing line, as the positive direction ratio tends to be larger than the negative direction ratio in stock markets. In Fig 11, the colored d_L points tend to be located closest to the blue stars, mostly in order by profit: red is the closest, followed by pink and orange. This result suggests that *trend strength* and the *trend direction ratio* are the key factors



Fig 11. Averaging windows and the balance between positive and negative market directions. The red, pink, and orange figures are the first, second, and third profit-maximizing d_L values, respectively. The blue star at the upper right end of the black upward-sloping line indicates the direction for maximizing profit while keeping the upward/downward ratio of the corresponding market.

to consider when selecting the return-maximizing d_L best adapted to occasional trend direction phase changes; that is, non-stationarity.

We suggest the mean returns of the target price series as a simple guideline to select the optimal d_L . The mean returns reflect the key information of both trend strength and the trend direction ratio. A stronger trend yields a larger mean return. The correlation between the monthly positive price direction ratio and mean returns in the same year is around 80% for each market. Fig 12 illustrates the relationship between the mean returns and the return-maximizing d_L value. We divide the 25 out-of-sample test years into five blocks (1995-1999, 2000-2004, 2005-2009, 2010-2014, and 2015-2019) and plot the mean returns during each block and the corresponding profit-maximizing d_L value. The orange, red, light blue, and blue points indicate the NYSE, NASDAQ, FTSEALL, and TOPIX, respectively. We find a positive relationship between mean returns and the profit-maximizing d_L value. The estimated coefficients of the simple regression lines in each panel are all significant, with p-values less than 1%.

Using this relation, we can compute a roughly appropriate d_L depending on the market condition, namely trend strength and the trend direction ratio. In the case of monthly trading by LTDC, for example, an appropriate d_L to use is determined by $\hat{d}_L^* = 102.27 + 10781.68 \ \bar{r}_{5y}$, where \bar{r}_{5y} is the mean returns during the past five years. By incorporating this d_L determination





https://doi.org/10.1371/journal.pone.0276322.g012

step before SVM re-construction, we can implement LTDC trading with adaptive d_L conditional on market trend condition. Note that d_L must be determined *before* SVM re-construction, because d_L controls the target trend scale to classify. By executing the backtest, this method can be implemented to any trend-following rule to trade any price series. The realized past returns are a helpful first step in selecting the appropriate d_L value to maximize the trading performance of trend-following rules by keeping the best balance of "riding the upward trend" and "being cautious about occasional trend reversals".

For further improving the performance of return maximizing d_L estimation, it is important to examine the effect of CV block size to obtain the optimal d_L . Whereas our preliminary test results reveal that the effect from extending the training data period significantly exceeds that of changing the CV block size, it is desirable to thoroughly examine the direct effect of changing CV block size to the return maximizing dL determination.

6.4 Number of trades

The selection of d_L significantly affects the number of trades, and hence, transaction costs. We compute the trading results when the trade is executed only when the buy/sell decision changes from the last month. We depict the relationship between the number of transactions during the 25-year out-of-sample test period and d_L in Fig 13. The number of trades using the trend-following rules decreases as d_L increases because the target trend scale grows as d_L increases, reducing the number of trend reversals. While the number of trades declined significantly in the monthly trading experiment, the cumulative returns corresponding to Fig 13 improved slightly from the results in Fig 5. The trend-following rules' common optimum d_L during the 25-year test period is not less than 100, and the number of trend-following GSVM rule. The number of trades using the LTDC is usually the lowest for all d_L values and all indices.

7 Discussion

This article studies the problems that trend-following trading confronts, and proposes a simple data-based method of finding an appropriate averaging window which is the key control parameter common for trend-following rules. The limitation of this study is that we have not reached to incorporate our result to provide a practical online trading algorithm. To realize a practical online trading algorithm where traders enjoy the advantage of LTDC while keeping





the averaging window at optimal level at every trading decision, it is important to reduce the computational cost of LTDC. The solution key is to design the learning system customized to our specific problem, long-term trend direction classification, to make the simplest possible data-driven system [4-6].

The nature of change we need to handle is determined by our aim, avoiding losses from long-term trend reversals. Hence LTDC requires the longest available training data period to learn the comprehensive relationship between return density shape (via moment statistics) and long-term trend direction including rare large drawdown. Learning longest available data period affects the design of the learning system in two ways: 1) it makes difficult to reduce computational cost of model construction, and 2) it makes the durability of constructed model longer and reduces the gain from re-construction as training period increases.

To reduce the computational cost of windowing techniques in online concept drift learning literature, most approaches aim at reducing the training data such as "forget memory (discard old training data)" or "discard irrelevant training data" [30, 37]. The key for reducing the training cost without shortening training data period, therefore, becomes discriminating the essential information from the training data. Extracting information necessary to learn the cases that are easy to mis-classify seems prominent, but immediate implementation is difficult since it requires further studies on the essential affecting factors to change long-term trend directions.

As to the model durability, there are studies reporting that the classification performance of well pre-trained learning machine based on cross-validation can keep offering better solution than that of online dynamic approach [59, 60]. The SVM of LTDC learns the long-term trend direction in relation with the density shape expressed by return moment statistics, and the extra gain of model re-construction by adding new data decreases as the training data period becomes longer. If the initial learning based on the longest available data has sufficient quality, frequent model re-construction might not be required by implementing some fine-tuning algorithm in generating decision.

Given above discussion, a possible solution is a system combining full-learning model and detection strategy: 1) multiple model (re-)construction with different averaging window (d_L) settings by SVM in LTDC framework, and 2) online optimum d_L model selection based on the latest past mean returns. The relationship between return maximizing d_L and the latest past mean returns is estimated at the timing of model (re-)construction. Before every trade decision, the recent return moment statistics is input to the selected model using most appropriate

 d_L to generate a trade decision. Here, the trend strength (the status of non-stationarity) and trend direction ratio (the degree of class imbalance) is monitored via the recent mean return levels to keep selecting the return-maximizing d_L model.

Another direction to reduce computational cost is employing unsupervised learning which does not require training data. This study adopts supervised learning through k-fold CV since the current unsupervised learning yet remains placed as challenging approach especially in terms of the accuracy [61, 62], and cross-validation-selected algorithm is guaranteed as best-performing algorithm [60]. LTDC rule learns return density shape via moment statistics, therefore, it might also be a prominent direction to pursue unsupervised learning such as online density estimation exploiting more detailed information about the non-stationarity status.

8 Conclusions

This study confirms the advantage of trend-following trading in avoiding losses from longterm trend reversals, and empirically demonstrates that the long-term trend direction change (non-stationarity) and trend direction ratio (class imbalance) are the fundamental affecting factors to trend-following trading performance. The difference in the optimum averaging window, d_L , by trend direction phase is found to lead to the stability-plasticity dilemma: the tradeoff between "keep riding the upward price trend" and "be cautious about sudden trend reversals" in adapting to the time-varying trend reversal risks. That is, the dilemma of selecting a high d_L value for a stable identification of the upward trend direction, and a low d_L value for the sensitivity (plasticity) to capture trend reversal signals. We demonstrate several effective directions to improve conventional trend-following trading performance by outperformance of LTDC rule, and propose a data-based appropriate averaging window finding method adaptive to the market condition.

This study explains that the unstable profitability of trend-following is caused by insufficient chances of avoiding trend reversals. It provides insight to investors that trend-following strategies performs best relative to BHR in risky markets when downside risk protection is most needed. The proposed averaging window finding method is the first data-based guideline easily applicable in practical trading, which is realized through investigation of comprehensive relationship between the optimal averaging window size and various market conditions. It would be more helpful for investors than using some popular averaging window size. Moreover, we are the first to introduce machine classification of smoothed long-term trend direction to generate trading decision, paving the way to explore the significant improvement of long-term downside risk aversion ability.

The remaining problem is incorporating our result to provide the practical online trading algorithm. It is needed to evaluate the practical performance of the suggested learning system combining the full-learning LTDC model and online optimal d_L selection algorithm. Further understanding about long-term trend direction behavior is also the key for discriminating unnecessary instances from the training data to achieve computational cost reduction.

Predicting large long-term trend reversal is quite challenging, and few studies have tackled with it. Trend following is currently the only trading strategy that aims making profit by avoiding large losses from long-term trend reversals. Trend direction classification studies of machine-learning-based approaches have focused on short-term trend direction because of the challenging nature of long-term trend direction classification. By directly handling the essential difficulties lying behind it, non-stationarity with class imbalance, this study helps improve the performance of all trend-following rules, being beneficial to both academia and practitioners.

Acknowledgments

The authors greatly appreciate the referees for their valuable comments, which substantially improved this manuscript.

Author Contributions

Conceptualization: Teruko Takada.

Data curation: Takahiro Kitajima.

Formal analysis: Teruko Takada, Takahiro Kitajima.

Funding acquisition: Teruko Takada.

Investigation: Teruko Takada, Takahiro Kitajima.

Methodology: Teruko Takada, Takahiro Kitajima.

Project administration: Teruko Takada.

Resources: Teruko Takada.

Software: Teruko Takada, Takahiro Kitajima.

Supervision: Teruko Takada.

Validation: Teruko Takada, Takahiro Kitajima.

Visualization: Teruko Takada.

Writing – original draft: Teruko Takada.

Writing - review & editing: Teruko Takada, Takahiro Kitajima.

References

- 1. Grossberg S. Nonlinear neural networks: Principles, mechanisms, and architectures. Neural Networks. 1988; 1(1):17–61. https://doi.org/10.1016/0893-6080(88)90021-4
- 2. Yang Q, Wu X. 10 challenging problems in data mining research. International Journal of Information Technology & Decision Making. 2006; 05(04):597–604. https://doi.org/10.1142/S0219622006002258
- Elwell R, Polikar R. Incremental learning of concept drift in nonstationary environments. IEEE Transactions on Neural Networks. 2011; 22(10):1517–1531. https://doi.org/10.1109/TNN.2011.2160459 PMID: 21824845
- 4. Žliobaitė I. Learning under concept drift: An overview. arXiv:10104784. 2010; p. 1–36.
- Hoens TR, Polikar R, Chawla NV. Learning from streaming data with concept drift and imbalance: An overview. Progress in Artificial Intelligence. 2012; 1:89–101. https://doi.org/10.1007/s13748-011-0008-0
- Žliobaitė I, Pechenizkiy M, Gama J. An overview of concept drift applications. In: Japkowicz N, Stefanowski J, editors. Big Data Analysis: New Algorithms for a New Society. Springer; 2016. p. 91–114.
- Wang S, Minku LL, Yao X. A systematic study of online class imbalance learning with concept drift. IEEE Transactions on Neural Networks and Learning Systems. 2018; 29(10):4802–4821. <u>https://doi.org/10.1109/TNNLS.2017.2771290</u> PMID: 29993955
- Cano A, Krawczyk B. Kappa updated ensemble for drifting data stream mining. Machine Learning. 2020; 109:175–218. https://doi.org/10.1007/s10994-019-05840-z
- Brock W, Lakonishok J, LeBaron B. Simple technical trading rules and the stochastic properties of stock returns. The Journal of Finance. 1992; 47(5):1731–1764. https://doi.org/10.1111/j.1540-6261.1992. tb04681.x
- Szakmary AC, Shen Q, Sharma SC. Trend-following trading strategies in commodity futures: A reexamination. Journal of Banking & Finance. 2010; 34(2):409–426. https://doi.org/10.1016/j.jbankfin. 2009.08.004

- Moskowitz TJ, Ooi YH, Pedersen LH. Time series momentum. Journal of Financial Economics. 2012; 104(2):228–250. https://doi.org/10.1016/j.jfineco.2011.11.003
- Neely CJ, Rapach DE, Tu J, Zhou G. Forecasting the equity risk premium: The role of technical indicators. Management Science. 2014; 60(7):1772–1791. https://doi.org/10.1287/mnsc.2013.1838
- 13. Marshall BR, Nguyen NH, Visaltanachoti N. Time series momentum and moving average trading rules. Quantitative Finance. 2017; 17(3):405–421. https://doi.org/10.1080/14697688.2016.1205209
- 14. Faber M. A quantitative approach to tactical asset allocation revisited 10 years later. The Journal of Portfolio Management. 2018; 44(2):156–167. https://doi.org/10.3905/jpm.2018.44.2.156
- 15. Daniel K, Moskowitz TJ. Momentum crashes. Journal of Financial Economics. 2016; 122(2):221–247. https://doi.org/10.1016/j.jfineco.2015.12.002
- Hurst B, Ooi YH, Pedersen LH. A century of evidence on trend-following investing. The Journal of Portfolio Management. 2017; 44(1):15–29. https://doi.org/10.3905/jpm.2017.44.1.015
- 17. Zhang XF. Information uncertainty and stock returns. The Journal of Finance. 2006; 61(1):105–137. https://doi.org/10.1111/j.1540-6261.2006.00831.x
- Han Y, Yang K, Zhou G. A new anomaly: The cross-sectional profitability of technical analysis. Journal of Financial and Quantitative Analysis. 2013; 48(5):1433–1461. <u>https://doi.org/10.1017/</u> S0022109013000586
- Clare A, Seaton J, Smith PN, Thomas S. Trend following, risk parity and momentum in commodity futures. International Review of Financial Analysis. 2014; 31:1–12. https://doi.org/10.1016/j.irfa.2013. 10.001
- Bajgrowicz P, Scaillet O. Technical trading revisited: False discoveries, persistence tests, and transaction costs. Journal of Financial Economics. 2012; 106(3):473–491. <u>https://doi.org/10.1016/j.jfineco.2012.06.001</u>
- 21. Fang J, Jacobsen B, Qin Y. Predictability of the simple technical trading rules: An out-of-sample test. Review of Financial Economics. 2014; 23(1):30–45. https://doi.org/10.1016/j.rfe.2013.05.004
- 22. Szakmary AC, Lancaster MC. Trend-following trading strategies in U.S. stocks: A revisit. Financial Review. 2015; 50(2):221–255. https://doi.org/10.1111/fire.12065
- 23. Strobel M, Auer BR. Does the predictive power of variable moving average rules vanish over time and can we explain such tendencies? International Review of Economics & Finance. 2018; 53:168–184. https://doi.org/10.1016/j.iref.2017.10.012
- 24. Olson D. Have trading rule profits in the currency markets declined over time? Journal of Banking & Finance. 2004; 28(1):85–105. https://doi.org/10.1016/S0378-4266(02)00399-0
- Auer BR. Have trend-following signals in commodity futures markets become less reliable in recent years? Financial Markets and Portfolio Management. 2021; 35:533–553. <u>https://doi.org/10.1007/s11408-021-00385-5</u>
- Chordia T, Subrahmanyam A, Tong Q. Have capital market anomalies attenuated in the recent era of high liquidity and trading activity? Journal of Accounting and Economics. 2014; 58(1):41–58. <u>https://doi.org/10.1016/j.jacceco.2014.06.001</u>
- 27. McLean RD, Pontiff J. Does academic research destroy stock return predictability? The Journal of Finance. 2016; 71(1):5–32. https://doi.org/10.1111/jofi.12365
- Ferreira FF, Silva AC, Yen JY. Detailed study of a moving average trading rule. Quantitative Finance. 2018; 18(9):1599–1617. https://doi.org/10.1080/14697688.2017.1417621
- 29. Zakamulin V, Giner J. Trend following with momentum versus moving averages: A tale of differences. Quantitative Finance. 2020; 20(6):985–1007. https://doi.org/10.1080/14697688.2020.1716057
- Klinkenberg R, Joachims T. Detecting concept drift with support vector machines. In: Proceedings of the Seventeenth International Conference on Machine Learning; 2000. p. 487–494.
- **31.** Bifet A, de Francisci Morales G, Read J, Holmes G, Pfahringer B. Efficient online evaluation of big data stream classifiers. In: Proceedings of the 21th ACM SIGKDD International Conference on Knowledge Discovery and Data Mining; 2015. p. 59–68.
- Brzezinski D, Stefanowski J. Prequential AUC: Properties of the area under the ROC curve for data streams with concept drift. Knowledge and Information Systems. 2017; 52:531–562. https://doi.org/10. 1007/s10115-017-1022-8
- Gama J, Žliobaitė I, Bifet A, Pechenizkiy M, Bouchachia A. A survey on concept drift adaptation. ACM Computing Surveys. 2014; 46(4):1–37. https://doi.org/10.1145/2523813
- Chen J, Hong H. Discussion of "Momentum and autocorrelation in stock returns". The Review of Financial Studies. 2002; 15(2):565–574. https://doi.org/10.1093/rfs/15.2.565
- Jegadeesh N, Titman S. Returns to buying winners and selling losers: Implications for stock market efficiency. The Journal of Finance. 1993; 48(1):65–91. https://doi.org/10.1111/j.1540-6261.1993.tb04702.x

- 36. Vapnik V. The Nature of Statistical Learning Theory. Berlin, Springer; 1995.
- Iwashita AS, Papa JP. An overview on concept drift learning. IEEE Access. 2019; 7:1532–1547. https:// doi.org/10.1109/ACCESS.2018.2886026
- Huang W, Nakamori Y, Wang SY. Forecasting stock market movement direction with support vector machine. Computers & Operations Research. 2005; 32(10):2513–2522. <u>https://doi.org/10.1016/j.cor.</u> 2004.03.016
- Kim Kj. Financial time series forecasting using support vector machines. Neurocomputing. 2003; 55(1-2):307–319. https://doi.org/10.1016/S0925-2312(03)00372-2
- 40. Kumar M, Thenmozhi M. Forecasting stock index movement: A comparison of support vector machines and random forest. In: Indian Institute of Capital Markets 9th Capital Markets Conference Paper; 2006. p. 1–16.
- Lee MC. Using support vector machine with a hybrid feature selection method to the stock trend prediction. Expert Systems with Applications. 2009; 36(8):10896–10904. <u>https://doi.org/10.1016/j.eswa.2009</u>. 02.038
- Kara Y, Boyacioglu MA, Baykan ÖK. Predicting direction of stock price index movement using artificial neural networks and support vector machines: The sample of the Istanbul Stock Exchange. Expert Systems with Applications. 2011; 38(5):5311–5319. <u>https://doi.org/10.1016/j.eswa.2010.10.027</u>
- **43.** Nadaraya EA. On estimating regression. Theory of Probability & Its Applications. 1964; 9(1):141–142. https://doi.org/10.1137/1109020
- Amaya D, Christoffersen P, Jacobs K, Vasquez A. Does realized skewness predict the cross-section of equity returns? Journal of Financial Economics. 2015; 118(1):135–167. <u>https://doi.org/10.1016/j.jfineco.2015.02.009</u>
- Scheffer M, Bascompte J, Brock WA, Brovkin V, Carpenter SR, Dakos V, et al. Early-warning signals for critical transitions. Nature. 2009; 461:53–59. https://doi.org/10.1038/nature08227 PMID: 19727193
- Scheffer M, Carpenter SR, Lenton TM, Bascompte J, Brock W, Dakos V, et al. Anticipating critical transitions. Science. 2012; 338(6105):344–348. https://doi.org/10.1126/science.1225244 PMID: 23087241
- Scheffer M, Carpenter SR, Dakos V, van Nes EH. Generic indicators of ecological resilience: Inferring the chance of a critical transition. Annual Review of Ecology, Evolution, and Systematics. 2015; 46:145–167. https://doi.org/10.1146/annurev-ecolsys-112414-054242
- Bergmeir C, Benítez JM. On the use of cross-validation for time series predictor evaluation. Information Sciences. 2012; 191:192–213. https://doi.org/10.1016/j.ins.2011.12.028
- Bergmeir C, Costantini M, Benítez JM. On the usefulness of cross-validation for directional forecast evaluation. Computational Statistics & Data Analysis. 2014; 76:132–143. <u>https://doi.org/10.1016/j.csda.</u> 2014.02.001
- Agrahari S, Singh AK. Concept drift detection in data stream mining: A literature review. Journal of King Saud University-Computer and Information Sciences. 2021; <u>https://doi.org/10.1016/j.jksuci.2021.11</u>. 006
- Tversky A, Kahneman D. Advances in prospect theory: Cumulative representation of uncertainty. Journal of Risk and Uncertainty. 1992; 5:297–323. https://doi.org/10.1007/BF00122574
- Reschenhofer E, Sinkovics T. Examining the profitability of automatic trading strategies with a focus on trend indicators. Quantitative Finance. 2017; 17(7):979–991. https://doi.org/10.1080/14697688.2016. 1262055
- Karatzoglou A, Smola A, Hornik K, Zeileis A. kernlab-an S4 package for kernel methods in R. Journal of Statistical Software. 2004; 11(9):1–20. https://doi.org/10.18637/jss.v011.i09
- Chang CC, Lin CJ. LIBSVM: A library for support vector machines. ACM Transactions on Intelligent Systems and Technology. 2011; 2(3):1–27. https://doi.org/10.1145/1961189.1961199
- 55. White H. A reality check for data snooping. Econometrica. 2000; 68(5):1097–1126. https://doi.org/10. 1111/1468-0262.00152
- Politis DN, Romano JP. The stationary bootstrap. Journal of the American Statistical Association. 1994; 89(428):1303–1313. https://doi.org/10.1080/01621459.1994.10476870
- Patton A, Politis DN, White H. Correction to "Automatic block-length selection for the dependent bootstrap" by D. Politis and H. White. Econometric Reviews. 2009; 28(4):372–375. https://doi.org/10.1080/ 07474930802459016
- 58. Kim AY, Tse Y, Wald JK. Time series momentum and volatility scaling. Journal of Financial Markets. 2016; 30:103–124. https://doi.org/10.1016/j.finmar.2016.05.003
- Hashmani MA, Jameel SM, Alhussain H, Rehman M, Budiman A. Accuracy performance degradation in image classification models due to concept drift. International Journal of Advanced Computer Science and Applications. 2019; 10(5):422–425. https://doi.org/10.14569/IJACSA.2019.0100552

- 60. Benkeser D, Ju C, Lendle S, van der Laan M. Online cross-validation-based ensemble learning. Statistics in Medicine. 2018; 37(2):249–260. https://doi.org/10.1002/sim.7320 PMID: 28474419
- Lu J, Liu A, Dong F, Gu F, Gama J, Zhang G. Learning under concept drift: A review. IEEE Transactions on Knowledge and Data Engineering. 2019; 31(12):2346–2363. <u>https://doi.org/10.1109/TKDE.2018</u>. 2876857
- **62.** Hoi SCH, Sahoo D, Lu J, Zhao P. Online learning: A comprehensive survey. Neurocomputing. 2021; 459:249–289. https://doi.org/10.1016/j.neucom.2021.04.112